
CENWP-OP-B                                                                                                                                  10 June, 2002

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 23 of 2002, which covers the period 02 through 08 June 2002.

1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 253.2 to 375.2kcfs.  Daily project tailwater elevation averaged 23.2’ to 27.0’ msl.  Daily project forebay elevation averaged 74.2’ to 75.9’ msl.  River temperature averaged 56 to 57° F.  Daily turbidity measured 2.8 to 4.0’ on the secchi disk.

1.2 Spill Bays 1and 18, which supply attraction flow for Cascades Island and Bradford Island ladders respectively, are operational and included in the spill pattern.  Spill averaged 144.4 to 208.0kcfs for the week.  

1.3 Unit Operation

1.3.1 Unit 2 returned to service at 1900 on 6 June. 

1.3.2 Unit 5 remains out of service for turbine rehab.  

1.3.3 Unit 9 remains out of service for digital governor work and a five year overhaul.

1.3.4 Unit 15 remains out of service until the VBSs are repaired.

1.3.5 Unit 17 was out of service for VBS cleaning from 0818 until 1200 on 3 June, 0742 until 0954 on 6 June and from 0718 until 1318 on 8 June.  Unit 17 was also out of service for STS repair from 0740 until 1542 on 4 June.

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.  Both fish units were on standby from 1230 on 02 June until 1006 on 03 June because of  the Washington Shore exit log boom failure (see misc. 5.5).  FU1 was out of service due to a clogged cooling water strainer from 1400 until 1506 on 06 June.

1.5 Gatewell drawdowns were measured every day for Unit 17 and on 5 June for the rest of the units.

2.  ACTIVITIES. 
2.1 AWS Closures.  The following AWS valves were closed this reporting week.

02 June- FV5-3, FV5-4, FV5-9, and FV6-9 were closed from 1242 on 02 June until 1006 on 03 June due to the broken log boom at the Washington Shore exit.

06 June- FV3-9 closed for cleaning from 1510 until 1600.

2.2 STS/VBS Inspections.  No STS/VBSs were inspected this reporting week.

2.3 Dewatering and Fish Salvage.  Biologists recovered two 2.5’ sturgeon, approx. 20 juvenile steelhead and approx. 10 juvenile Chinook from DSM1 on 07 June.  All fish were in good condition except for 5 chinook which were dead.  

3.  RESEARCH.

3.1 CRITFC Age and Length Composition.  Collected 40 Spring Chinook this reporting week.

3.2 NMFS FGE.  Unit 17 FGE study is on a week hiatus.

3.3 NMFS Lamprey.  Collection continues.  Fish are being collected for NMFS, UI, USGS and the Warm Springs tribe.

3.4 NMFS Saltwater Rearing.  Collection and rearing continue. 

3.5 Normandeau Balloon-tag at Spillway.  High flow tests ended on 1 June.  Low flow study commences in  August.

3.6 OCFWRU Transportation Study.  Study of post-transportation migration and survival of juveniles to run 29 April through 14 July.

3.7 ODFW Summer Steelhead sampling.  Sampling 2 days/week 8 hours/day.  Will sample 1% of the run.

3.8 Pelfrey Sturgeon Hatchery.  Capturing broodstock May 24 – June 24.

3.9 PSMFC Adult PIT.  Detectors are installed in all ladders.  Some of the detectors have developed leaks and will be monitored throughout the season.  Repairs are scheduled for 2002-03 winter maintenance season.

3.10 PSMFC JMF.  Three kelts (2 alive and 1 dead) were removed from the separator bars.  Fallbacks included 246 steelhead, 32 salmon and 11 unknown salmonids; 895 smolts slid over the separator bars as well.

3.11 PSMFC/NMFS Juvenile PIT.  91 yearling Chinook and 80 steelhead were examined for signs of GBT.  Total descaling for fish sampled at PH1 was 14.2%, similar to 14.5% last week.  Excessive descaling was noted early in the week (25% overall) but tapered off (10%).  Similar descaling trends were observed last week.  See misc 5.4
3.12 U of I Adult Radio tracking.  Trapping began 1 April and will continue 7 days/week until October.

3.13 U of I Transportation Study.  Evaluation of physiological condition of river run v. barged Snake River juveniles to occur June-September.

3.14 USGS EMG studies.  Tissue samples will be taken from 200 Chinook 1 April through June.

3.15 USGS Juvenile Radio tracking.  Study in progress.

3.16 USGS Lamprey.  Collection of lamprey at the AFF began in May.

3.17 USGS MGR.  Released fish nearly every day this week.  Last release scheduled for 10 June. 

3.18 USGS Sort by Code.  Length and weight are recorded and fish are released back to the river.

3.19 WDFW Fish Counting.  16-hr visual and 8-hr video counts at Bradford Island and Washington Shore.

3.20 WES/Battelle/Mevatech Juvenile Hydroacoustics.  Study in progress.  

3.21 WES Spillway Deflector Testing.  WES monitoring water quality equipment to determine effectiveness of new flip lips.  Study began with initiation of spill on 10 April.

4.  FISHWAYS 
4.1 Project biologists inspected 03, 04 and 06 June this reporting week.

4.2 Fish Passage Plan Observations.

	DATE
	LOCATion
	FPP Deviation
	CAUSE
	RESPONSE
	TR/WAR

	All week
	PH1 PCC velocity
	Velocity <1.5fps on the electronic meter.
	Unknown- the rest of the ladder is in criteria.
	No action at this time.
	

	6/2, 6/5, 6/7-8
	
	Biologists did not inspect 7 days/ week.
	Personnel shortage.
	Job advertised.
	

	6/6
	PH1 PCC
	OG#9 not at correct elevation.
	Unknown.
	Gate manually adjusted.
	

	6/3
	PH1 PCC
	WG#1 provided no entrance.
	Maintaining differential instead of opening.
	WG1 and FV1-1 in manual and adjusted.  (see misc 5.6)
	TR#7723

	6/3
	Washington Shore
	Ladder around orifice flow
	Log boom broke, debris piled on exit trash rack.
	Ladder shut down until riggers could secure log boom and remove debris.
	TR#7513

	All week
	CI and B-Branch 
	Diffusers not in correct position for TW elevation.
	Operators manually adjusted diffusers.
	See Misc. 5.1
	


4.3 Fish Ladder Temperatures.  Temperatures are measured 3 weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.  Temperatures are in plain old Celsius this week.  

	Date
	Wash. Shore
	Bradford Is.

	6/02
	14.5
	14.6

	6/03
	14.7
	14.7

	6/04
	14.9
	14.9

	6/05
	15.0
	15.0

	6/06
	14.6
	14.6

	6/07
	14.3
	14.3

	6/08
	14.3
	14.3


4.4 Sensor Calibration.  Calibration was checked on 06 June this reporting week.  Everything is within criteria.

4.5 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week. 

5. MISCELLANEOUS

5.1 B-branch and Cascade Island are operated in manual mode until stillwells can be installed for the tailwater sensors.  Installation will wait until spill has concluded.

5.2 Turbulent flows from spillbays 1 and 18 enter the side entrance gates of Cascade Island and Bradford Island B-branch ladders, causing poor entrance conditions.  District hydraulic engineers are investigating the problem.  On 3 May, the structural crew installed a bulkhead in the side entrance at Cascade Island.  Entrance conditions improved.  CG funds will be used to build a bulkhead for Bradford Island B-branch.

5.3 Juvenile mortalities at the JMF continue to be an intermittent problem.  Mort. numbers are down since last week.  PSMFC is digitally photographing some of the fish.  DSM2 Orifices are flushing six times a day.

5.4 Pacific States reported that intermittent descaling persisted at DSM1 this week although the problem was less severe than it was last week.  Orifices are flushing six times a day.  Project inspected screens and raked trashracks, discovering  nothing unusual.

5.5 At 1220 on 02 June the Washington Shore fish exit log boom broke.  Debris on the trashracks impaired fish passage.  No maintenance crews were available to repair the boom, operators proceeded to shut down the ladder to orifice flow.  Fish units went to standby at 1230, AWS valves were closed and by 1340 both Cascade Island and Washington Shore were at orifice flow.  PH1became the priority powerhouse.  At 1006 on 03 June, the log boom was secure, the ladders were back in normal operation and operating priority was back at PH2.

5.6 There has been some difficulty in maintaining an opening for fish at PH1 weir gate 1.  The gate is currently programmed not to close if there is less than an 8’ opening, but it also won’t open automatically unless there is a problem with differential.  Electricians are exploring the possibility of reprogramming the gate such that it will open if there is less than a 6’ opening and not close if less than 8’.  Until then, both WG1 and FV1-1 will be manually adjusted as necessary to keep an opening at the south end of PH1 collection channel.

5.7 FV6-9 was in manual from 1235 until 1602 on 03 June while a new sonic transducer was installed.

5.8 The DSM1 was turned south from 0820 until 1330 on 07 June to facilitate repairs to the small trashsweep.

5.9 Gatewell dipping for Unit 17 VBS cleanings are ongoing.  VBSs were cleaned on 03, 06 and 08 June.  On 08 June, the A-slot VBS was damaged and needed to be replaced.

 DEBORAH L. CHENOWETH

                                 

                                                             Operations Project Manager
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